Author's response to reviews

Title: Study protocol of cost-effectiveness and cost-utility of a biopsychosocial multidisciplinary intervention in the evolution of non-specific sub-acute low back pain in the working population: cluster randomised trial

Authors:

Anna Berenguera (aberenguera@idiapigol.org)
Enriqueta Pujol-Ribera (epujol@idiapigol.org)
Teresa Rodriguez-Blanco (trodriguez@idiapigol.org)
Concepción Violan (cviolan@idiapigol.org)
Marc Casajuana (mcasajuana@idiapigol.org)
Nelleke DeKort (dekortnelleke@gmail.com)
Marta Trapero-Bertran (marta.trapero-bertran@brunel.ac.uk)

Version: 2 Date: 27 July 2011

Author's response to reviews:

Dear Editor,

We have made some errors during the submission process.

Please find attached our manuscript entitled “Study protocol of cost-effectiveness and cost-utility of a biopsychosocial multidisciplinary intervention in the evolution of non-specific sub-acute low back pain in the working population: cluster randomised trial” which we would like to be considered for publication in the BMC Musculoskeletal Disorders.

Manuscript type: Study Protocol

This manuscript deals with a relevant project that we would like to contribute evidence about the cost-effectiveness and cost-utility of multidisciplinary biopsychosocial educational group intervention in reducing episodes of non-specific Low Back pain and the associated social and health in a costs. This evidence could help health professionals and administrators to make more efficient decisions to address this problem in the primary care setting. Furthermore, in this study, it will be adopted a social perspective; it is unusual to encounter economic evaluation studies with this perspective, even though low back pain is a disease with an important impact on the daily life of the patient as well as the society.

On behalf of the co-authors, I affirm that the manuscript, or any part of it, has neither been published nor is currently under consideration for publication by any other journal.

The current trials registration is in process. Application form CCT-NAPN-21250
All authors have read and approved the current manuscript.

We look forward to your opinion as to the suitability of our manuscript for inclusion in BMC Musculoskeletal Disorders.

Yours sincerely,

Enriqueta Pujol-Ribera
IDIAP Jordi Gol – Primary Health Care Institute
Research Department
Gran Via Corts Catalanes, 587 àtic
08007 – BARCELONA (SPAIN)
Tel. 0034 93 482 4342
Fax. 0034 93 482 4174
e-mail: epujol@idiapjgol.org